Author's response to reviews

Title: Treatment of Myofascial Trigger Points in Patients with Chronic Shoulder Pain: A Randomized Controlled Trial

Authors:

Carel Bron (carelbron@mac.com)
Arthur de Gast (adgast@diakhuis.nl)
Jan Dommerholt (dommerholt@bethesdaphysiocare.com)
Boudewijn Stegenga (b.stegenga@kchir.umcg.nl)
Michel Wensing (m.wensing@iq.umcn.nl)
Rob A.B. Oostendorp (r.oostendorp@iq.umcn.nl)

Version: 5 Date: 8 January 2011

Author's response to reviews: see over
Januari 8, 2011

Editor-in-chief
BMC Medicine

We would like to submit the revised version of the manuscript “Treatment of Myofascial Trigger Points in Patients with Chronic Shoulder Pain; A Randomized Controlled Trial” for publication in BMC Medicine. We are very pleased to know that you have accepted the manuscript for publication.

All requested corrections have been made.

- References: We have deleted reference 26, and instead we refer to it in the text according your recommendations (Authors: Title of manuscript, submitted).
- Figures 1a, b, c, and d and 2a, b, and c have been submitted as 1 to 7.

Yours sincerely,

Carel Bron, PT, MPT (PhD student)

Co-Authors: Arthur de Gast, MD, PhD; Jan Dommerholt PT, DPT; Boudewijn Stegenga DDS, PhD; Michel Wensing PhD Habil; Rob Oostendorp PT, MPT, PhD.